Author's response to reviews

Title: Detection of Norovirus genogroup I and II by multiplex real-time RT-PCR using a 3'-minor groove binder-DNA probe

Authors:

Marina Hoehne (hoehnem@rki.de)
Eckart Schreier (schreiere@rki.de)

Version: 7 Date: 7 April 2006

Author's response to reviews:

Article title: Detection of Norovirus genogroup I and II by multiplex real-time RT-PCR using a 3'-minor groove binder DNA probe

MS ID : 5665028184518471

Authors : Marina Hoehne and Eckart Schreier

Journal : BMC Infectious Diseases

The following formatting changes have been made:

- the abstract was already structured into Background, Method, Result, Conclusion, so I could not make any changes
- the titles included in the figures have been removed
- the figures saved in word have been cropped according to your cropping guidelines

Yours faithfully

Marina Hohne